

# 

**Citation:** Ding XC, Ade MP, Baird JK, Cheng Q, Cunningham J, Dhorda M, et al. (2017) Defining the next generation of *Plasmodium vivax* diagnostic tests for control and elimination: Target product profiles. PLoS Negl Trop Dis 11(4): e0005516. https://doi.org/10.1371/journal. pntd.0005516

Editor: Paulo Filemon Pimenta, Fundaçao Oswaldo Cruz, BRAZIL

Received: November 11, 2016

Accepted: March 21, 2017

Published: April 3, 2017

**Copyright:** This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

**Data Availability Statement:** All relevant data are within the paper and its Supporting Information files.

**Funding:** This work was supported by funds from the Department of Foreign Affairs and Trade, Australian Government. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

RESEARCH ARTICLE

# Defining the next generation of *Plasmodium vivax* diagnostic tests for control and elimination: Target product profiles

Xavier C. Ding<sup>1</sup>\*, Maria Paz Ade<sup>2</sup>, J. Kevin Baird<sup>3,4</sup>, Qin Cheng<sup>5</sup>, Jane Cunningham<sup>6</sup>, Mehul Dhorda<sup>7,8</sup>, Chris Drakeley<sup>9</sup>, Ingrid Felger<sup>10,11</sup>, Dionicia Gamboa<sup>12,13</sup>, Matthias Harbers<sup>14</sup>, Socrates Herrera<sup>15</sup>, Naomi Lucchi<sup>16</sup>, Alfredo Mayor<sup>17,18</sup>, Ivo Mueller<sup>19,20</sup>, Jetsumon Sattabongkot<sup>21</sup>, Arsène Ratsimbason<sup>22,23</sup>, Jack Richards<sup>24,25</sup>, Marcel Tanner<sup>10,11</sup>, Iveth J. González<sup>1</sup>

1 FIND, Geneva, Switzerland, 2 Department of Communicable Diseases and Health Analysis, Pan American Health Organization/World Health Organization, Washington, DC, Unites States of America, 3 Eijkman-Oxford Clinical Research Unit, Jakarta, Indonesia, 4 The Centre for Tropical Medicine, Nuffield Department of Medicine, University of Oxford, Oxford, United Kingdom, 5 Australian Army Malaria Institute, Brisbane, Australia, 6 Global Malaria Programme, World Health Organization, Geneva, Switzerland, 7 Centre for Tropical Medicine and Global Health, Nuffield Department of Medicine, University of Oxford, Oxford, United Kingdom, 8 World Wide Antimalarial Resistance Network, Churchill Hospital, Oxford, United Kingdom, 9 Department of Immunology and Infection, London School of Hygiene & Tropical Medicine, London, United Kingdom, 10 University of Basel, Basel, Switzerland, 11 Swiss Tropical and Public Health Institute, Basel, Switzerland, 12 Institute of Tropical Medicine Alexander von Humboldt, Universidad Peruana Cayetano Heredia, Lima, Peru, 13 Departamento de Ciencias Celulares y Moleculares, Facultad de Ciencias y Filosofía, Universidad Peruana Cayetano Heredia, Lima, Peru, 14 RIKEN Center for Life Science Technologies, Division of Genomic Technologies, Yokohama, Japan, 15 Caucaseco Scientific Research Center, Cali, Colombia, 16 Malaria Branch, Division of Parasitic Diseases and Malaria, Centers for Disease Control and Prevention, Atlanta, Georgia, United States of America, 17 ISGlobal, Barcelona Ctr. Int. Health Res. (CRESIB), Hospital Clínic-Universitat de Barcelona, Barcelona, Spain, 18 Centro de Investigação em Saúde da Manhiça (CISM), Maputo, Mozambique, 19 Population Health and Immunity Division, Walter and Eliza Hall Institute, Parkville, Australia, 20 Department of Parasites and Insect Vectors, Institut Pasteur, Paris, France, 21 Mahidol Vivax Research Unit, Faculty of Tropical Medicine, Mahidol University, Bangkok, Thailand, 22 Programme National de Lutte contre le Paludisme, Antananarivo, Madagascar, 23 University of Antananarivo, Antananarivo, Madagascar, 24 Centre for Biomedical Research, Burnet Institute, Melbourne, Victoria, Australia, 25 Department of Medicine, University of Melbourne, Parkville, Victoria, Australia

\* xavier.ding@finddx.org

## Abstract

The global prevalence of malaria has decreased over the past fifteen years, but similar gains have not been realized against *Plasmodium vivax* because this species is less responsive to conventional malaria control interventions aimed principally at *P. falciparum*. Approximately half of all malaria cases outside of Africa are caused by *P. vivax*. This species places dormant forms in human liver that cause repeated clinical attacks without involving another mosquito bite. The diagnosis of acute patent *P. vivax* malaria relies primarily on light microscopy. Specific rapid diagnostic tests exist but typically perform relatively poorly compared to those for *P. falciparum*. Better diagnostic tests are needed for *P. vivax*. To guide their development, FIND, in collaboration with *P. vivax* experts, identified the specific diagnostic needs associated with this species and defined a series of three distinct target product profiles, each aimed at a particular diagnostic application: (i) point-of-care of acutely ill patients for clinical care purposes; (ii) point-of-care asymptomatic and otherwise sub-

patent residents for public health purposes, e.g., mass screen and treat campaigns; and (iii) ultra-sensitive not point-of-care diagnosis for epidemiological research/surveillance purposes. This report presents and discusses the rationale for these *P. vivax*-specific diagnostic target product profiles. These contribute to the rational development of fit-for-purpose diagnostic tests suitable for the clinical management, control and elimination of *P. vivax* malaria.

### Author summary

Plasmodium vivax is the second most prevalent Plasmodium species amongst the five that can infect humans and cause malaria. The control and elimination of *P. vivax* is complicated by its specific biology, such as hard-to-detect low densities of blood-circulating parasites in infected individuals, the existence of persistent liver forms causing relapse, or the early appearance of sexual stages of the parasite during the course of an infection, which facilitates its transmission. These difficulties are reinforced by the fact that most antimalarial tools have been developed primarily for *P. falciparum*, the most prevalent malaria species, and are not always as effective for P. vivax. Current tools for the diagnosis of P. vivax are of limited effectiveness. Rapid diagnostic tests exist but show, in average, lower performance than similar test for P. falciparum. P. vivax diagnosis often relies on light microscopy which is challenging to maintain at a high quality and not sensitive enough to detect a large fraction of all infections. Recognizing that better diagnostic tools for P. vivax are needed, we report in this study the development of new target product profiles to define the specific characteristics of such tests. The establishment of these consensusbased documents is an important first step to guide research and development efforts toward better diagnostic solutions for P. vivax malaria and to accelerate the elimination of this species alongside P. falciparum.

### Introduction

#### The need for better Plasmodium vivax diagnostic tests

The concerted international efforts initiated near the turn of this century to move from malaria control to malaria elimination and ultimately eradication show remarkable progress during the last decade [1]. Financial, political and scientific commitment to solve the malaria problem led so far to an overall 37% decrease in global incidence between the years 2000 and 2015 and an estimated 60% decrease in mortality during this period [1]. As a result, in sub-Saharan Africa, where most known cases of malaria occur, malaria is no longer the prime cause of death for children below the age of 5 years old. These gains have been driven by the cumulative impacts of multiple entomological and antimalarial interventions implemented via improved policies. Since 2010 the World Health Organization (WHO), for example, recommended confirmation of suspected malaria cases using rapid diagnostic test (RDT) or by the examination of a stained blood smear by light microscopy (LM). The practice of presumptive treatment without confirmation was thus discouraged. Since then, the estimated number of malaria diagnostic tests performed globally on suspected cases has risen significantly, especially in the African WHO Region where the proportion of tested cases has increased from 41% in 2010 to 65% in 2014. This increase in testing is largely due to the availability of quality-assured RDTs. The number of RDT distributed by National Malaria Control Programs rose from less than 25

millions in 2008 to more than 125 millions in 2014 in the African WHO Region, whereas it remained relatively constant in other areas where *P. vivax* is present [1]. This highlights the importance of high quality, affordable and easy to use point-of-care tests to facilitate the effective diagnosis and prompt treatment of malaria.

This encouraging portrait of progress does not fully extend to the malaria caused by P. vivax. In fact, P. vivax prevalence appear to decrease slower than that of P. falciparum, which results in a significant shift to P. vivax predominance almost everywhere outside sub-Saharan Africa. P. vivax is now the sole or main cause of malaria in approximately three quarters (26/ 34) of the endemic countries currently in the elimination phase, suggesting that this species will be much more difficult to eliminate than P. falciparum. Although approximately one half (47%) of all malaria cases outside sub-Saharan Africa are caused by P. vivax and about 2.8 billion people live at risk of infection [2], it has been neglected in science, clinical medicine and public health until very recently [3–5]. This has resulted in strategies and tools for malaria control and elimination suited to P. falciparum but not P. vivax. Specific biological traits of P. vivax explain that poor fit. First, a single infectious bite of a mosquito leads to a primary attack within about 2 weeks, but then goes on to cause multiple clinical attacks at intervals of about 2 months for as long as 4 years (but typically 2 years or less). Those later attacks, called relapses, derive from dormant liver stages of *P. vivax* called hypnozoites. These forms have been shown to cause at least 80% of all *P. vivax* blood-stage infections in Papua New Guinea [6], and 96% of attacks at the Thai-Myanmar border [7]. Second, during the course of a blood stage infection, gametocytes appear simultaneously with the same 48 hour developmental cycle as asexual parasites and, often, before the onset of symptoms and, while they do not seem to persist as long as P. falciparum gametocytes, their infectiousness to mosquitoes may be relatively higher [8-11]. Third, P. vivax merozoites invade only the most immature reticulocytes that most often occur in bone marrow rather than in circulation [12]. The bulk of *P. vivax* biomass may occur in extravascular tissues of the marrow and spleen rather than in circulating blood, whereas *P. falciparum* is largely impounded within vascular sinuses [13]. Those observations help explain why parasitaemias by *P. vivax* are naturally and consistently much lower compared to P. falciparum. Parasite densities of P. vivax at clinical presentation are typically in the range of 4 000 +/- 3 000 parasite per  $\mu$ L of blood (p/ $\mu$ L), which is three to four-times lower than for *P. falciparum*, and peak parasitaemias rarely exceeds 100 000 p/µL in *P. vivax* but is quite common in *P. falciparum* [14–18]. In fact, a large proportion of all *P. vivax* infections, up to 70% in certain areas, have been found to be below the limit of detection (LOD) of microscopy [19]. Although present in all settings, submicroscopic infections appear to be clearly of higher relative importance in low prevalence areas, representing an additional challenge to elimination efforts [19].

The control and elimination of *P. vivax* is thus more complex than with *P. falciparum* as it requires the rapid diagnosis of infection at lower parasite densities, but also initiating radical cure treatment for hypnozoites in conjunction with acute treatment for blood stage parasites. Problematically, the only currently available hypnozoitocidal therapy, the 8-aminoquinoline primaquine regimen, is typically 14-days in duration and exposes glucose-6-phosphate dehydrogenase (G6PD) deficient patients–a widespread genetic disorder impacting 8% of residents of malaria endemic nations [20]–to potentially life-threatening acute haemolytic anaemia. Finally, some of the natural polymorphisms in the P450 cytochrome type 2D6 (CYP2D6) result in null or impaired metabolism of primaquine to its active metabolite and cause therapeutic failure against relapse [21–23].

The increasing use of RDTs for the diagnosis of malaria resulted in significant progress in the past ten years but less so for *P. vivax* specifically. Relatively poor diagnostic performance of the most widely used RDTs for non-*falciparum* malaria may help explain continuing reliance

upon microscopy for the diagnosis of *P. vivax* in endemic areas. RDTs for *P. vivax* are generally considered of lower accuracy, with performance, stability and false positivity issues being commonly reported in the literature [24-27]. The actual diagnostic coverage and the analytical performances of P. vivax RDTs are poorly documented. These deficiencies have been clearly recognised in the Malaria Eradication Research Agenda (malERA) initiative, which has expressed the high priority need for "more sensitive tests for *P. vivax* for case management" [28]. This initiative also highlighted that RDTs for P. vivax "lack consistency in sensitivity and stability" [28]. While a recent review has indicated that the quality of *P. vivax* RDTs is improving, only 59% (17/29) of P. vivax RDTs displayed an acceptable panel detection score (PDS) at 200 parasites per µl of blood as compared to 93% (38/41) for P. falciparum RDTs during the latest WHO-FIND product testing of malaria RDTs [25,29]. While P. vivax infections can be identified via the detection of *Plasmodium* specific aldolase or plasmodial lactate dehydrogenase (pLDH) enzymes, the univocal identification of *P. vivax* requires the specific detection of the pLDH isoform of this species (Pv-pLDH). As a proxy of the typical performances of this type of test, the average PDS of Pv-pLDH based RDTs appear significantly lower than that of the RDTs detecting P. falciparum specific histidine rich protein 2 (HRP2) when considering the cumulative results of the WHO-FIND Product Testing Programme (Table 1). This illustrates the shortcomings associated with current P. vivax specific RDTs.

Performance of light microscopy is directly dependent on operator proficiency and sample preparation, and species determination in areas of *P. falciparum* and *P. vivax* co-endemicity may be challenging [31,32]. Microscopy, like RDTs, also suffers limited sensitivity and often fails to identify a substantial fraction of *P. vivax* infections of blood [19]. Alternative diagnostic methods, based on nucleic acid amplification techniques (NAATs) and serological markers exist or are emerging. While microscopy and RDTs are recommended by WHO as "the primary diagnostic tools for the confirmation and management of suspected clinical malaria in all epidemiological situations including areas of low transmission as well as for routine malaria surveillance", a potential role for NAAT- and serology-based approaches is considered relevant in areas of low endemicity and near elimination for epidemiological research and surveys aimed at mapping submicroscopic infections to guide intervention measures specific to these settings [33]. While the research and laboratory applications of these tests is clear, their value for *P. vivax* infection detection and their optimal application are, however, currently unclear.

Currently available diagnostic tests for *P. vivax* are not optimal to address the full range of infection detection needs, from clinical case management to surveillance and elimination-oriented interventions through "surveillance-response". Currently, poor diagnostic effectiveness contributes to the resilience of *P. vivax* to global and national malaria intervention strategies.

Species	Antigen	n	Average PDS <sup>a</sup> and range		
P. vivax	Pv-pLDH	32	59% (0%-100%)		
P. vivax	Pvom-pLDH <sup>b</sup>	3	77% (63%-91%)		
P. vivax	aldolase	6	41% (0%-82%)		
P. falciparum	Pf-pLDH	9	52% (6%-89%)		
P. falciparum	HRP2	113	82% (32%-99%)		

Table 1. Average panel detection scores of quality-controlled RDTs from WHO-FIND Product Testing Programme (n = 126).

<sup>a</sup>Average panel detection score (PDS) of the corresponding *Plasmodium* species at 200 parasites per µL of blood. Extracted from [30].

<sup>b</sup>Representing the pLDH epitopes common to *P. vivax*, *P. ovale*, and *P. malariae*, enabling the indiscriminate detection of these three species.

https://doi.org/10.1371/journal.pntd.0005516.t001

### Defining target product profiles for Plasmodium vivax diagnostic tests

In order to facilitate the development of improved P. vivax tests, a set of target product profiles (TPPs) addressing the specific needs associated with this species were developed through expert consensus. These TPPs are intended to guide the efforts of test developers, donors and other stakeholders in the global health community to address the *P. vivax* challenge. A limited number of TPPs for malaria diagnostic tests have been developed in the past few years (S1 Table). The malERA initiative published two generic TPPs in 2011, one for the diagnosis of malaria clinical cases and one for screening and surveillance activities, with parameters that could be applied to both, P. falciparum and P. vivax [28]. In 2014, the 10th session of the WHO Malaria Policy Advisory Committee Meeting released recommendations for malaria diagnosis in low transmission settings and described the ideal characteristics of future tests for this application, without defining species-specific needs [34]. A malaria diagnostic TPP was also developed by PATH for the Diagnostics for Malaria Elimination Toward Eradication (DIAMETER) project that supports the development and implementation of diagnostic solutions for malaria elimination [35]. The format and target of the test described in this profile are restricted to that of lateral flow immunoassay detecting HRP2 for P. falciparum infections. Finally, while it did not include a full TPP, the 2015 WHO technical brief about the control and elimination of P. vivax malaria highlighted the need for research to develop tests that can detect P. vivax at a minimum of 25  $p/\mu L$  of blood as well as tests that "can detect submicroscopic, asymptomatic infections in elimination settings, where it is critical to detect all infections" [3]. While the very specific biological and clinical nature of P. vivax infections requires adapted tools, none of these TPPs addressed the needs of *P. vivax* infection detection. To fill this gap, FIND, a notfor-profit organization supporting the development and implementation of diagnostic solutions for diseases of poverty, consulted with P. vivax experts, all co-authors of this publication, to define the diagnostic needs for *P. vivax* and established consensus-based TPPs, with the goal to guide product development efforts toward optimized diagnostic solutions and to ultimately accelerate elimination of this malaria species.

### Methods

TTPs were developed in an iterative and consensus-decision-making process involving a large number of experts from academic research institutions, national malaria control programmes, the WHO Global Malaria Programme, and the WHO Americas Regional Office (AMRO)/Pan American Health Organization (PAHO). An initial expert meeting took place in October 2015 to review current practices and topics of interest for the diagnosis of *P. vivax* malaria. Three TPPs were defined based on specific intended uses, a list of forty-three TPP characteristics to be informed was established based on an initial list proposed by FIND, and preliminary values for each of these characteristics were discussed. TPPs were gradually refined through five rounds of drafts review and update through online communication (draft versions 0.1 to 0.6). Finally, an online survey was conducted to collect comments from each contributor on the remaining debated characteristics and establish a majority vote to issue final TPPs (versions 1.0) reported here.

For each TPP, the intended use, target populations and users, the implementation level as well as expected performance, operational and financial characteristics were defined. For most of these characteristics, minimal and optimal values have been defined, providing a range of values from the minimally acceptable value to the ideal one. The minimal values have been typically set to provide a distinguishing advantage over existing diagnostic solutions for *P. vivax* while the optimal ones were defined as the value that could provide optimal diagnostic effectiveness.

### Results

### Intended uses

The malaria diagnostic needs are wide and primarily defined by the type of infection to be detected, either restricted to clinical cases or including the largest possible number of infections, regardless of symptoms. Additional important factors are the test outcome, which can be to guide treatment or only to inform surveillance systems, and the implementation level, which will determine how simple to implement a given test needs to be. Three TPPs were defined to cover three distinct intended uses across this spectrum (Fig 1). TPP PvA (Pv stands for P. vivax) is addressing the diagnosis of P. vivax clinical symptomatic infection for confirmation of suspected cases (passive case detection). The two other TPPs (TPP PvB1 and PvB2) are geared toward elimination settings and address the diagnosis of all infections, symptomatic or not. TPP PvB1 addresses the need for point-of-care diagnosis of P. vivax infections regardless of the presence of symptoms including sub-microscopic parasitaemia, enabling proactive and reactive infection detection interventions, while TPP PvB2 specifically addresses the requirements for a population screening test for P. vivax infection surveillance and epidemiological surveys, independent of individual infection treatment. The specific intended uses and test outcomes as well as key distinguishing characteristics for these three TPPs are summarized in Table 2.

### TPP PvA: Diagnosis of P. vivax malaria acute infection

TPP PvA addresses the need for better diagnostic for the parasitological confirmation of clinical cases in passive case detection scenarios (S2 Table). This TPP is therefore designed for a point-of-care test that is simple to implement (requiring ideally half-a-day of training and



**Fig 1. Malaria testing strategies and TPP coverage.** Testing strategies are typically classified as passive and active detection where passive detection concerns symptomatic cases and active detection all infections, symptomatic or not. Passive detection is used for the confirmation of symptomatic suspected cases presenting to the healthcare system where treatment is based on a positive parasitological test (PCD: passive case detection). Active detection is typically divided as reactive and proactive detection where reactive detection consists of the active screening of a set of individuals linked geographically or sociologically to an index case for infection detection and treatment. Proactive detection can either be linked with treatment in focal screen-and-treat (FSAT) or mass screen-and-treat (MSAT) interventions or in location-based testing (*e.g.* boarder screening) or be independent of treatment in epidemiological surveys. The coverage of each of the three TPPs for *P. vivax* diagnostic tests, PvA, PvB1 and PvB2, is indicated in relation to these testing strategies. The classification of intervention types is adapted from [36].

https://doi.org/10.1371/journal.pntd.0005516.g001

three steps or less) and rapid (time-to-results  $\leq$  30 min.) to guide prompt clinical management of *P. vivax* malaria patient: blood-stage treatment of acute *P. vivax* infections as well as radical cure for populations to which 8-aminoquinolines can be administrated safely.

Туре	Characteristic	PvA	PvB1	PvB2
Scope	Intended use	For parasitological confirmation of symptomatic suspected cases of <i>P. vivax</i> malaria	For parasitological confirmation of all infections of <i>P. vivax</i> malaria (symptomatic and asymptomatic)	For indication of present or recent <i>P. vivax</i> infection for epidemiological surveys and surveillance activities
	Test outcome	Guide individual treatment in passive case detection	Guide individual treatment in reactive and proactive case detection	Inform epidemiological surveys, guide population interventions
	Target population	All individuals suspected to suffer from clinical <i>P. vivax</i> infection	All individuals susceptible to suffer from <i>P. vivax</i> infection in endemic settings	All individuals in an endemic setting
	Target users	M <sup>b</sup> : Community and facility- based health workers	M: Community and facility-based health workers	M: Laboratory technicians
	Implementation level	Community health facilities, health posts, health centers	Community health facilities, health posts, health centers	District hospitals and reference laboratories
Performance	Analytical sensitivity <sup>a</sup>	M: 25 p/μL Ο <sup>b</sup> : 5 p/μL	M: 20 p/μL Ο: 1 p/μL	M: 0.1 p/µL (irrelevant for recent past infection detection) O: 0.01 p/µL (irrelevant for recent past infection detection)
	Analytical specificity	M: Discriminate <i>P. vivax</i> form other <i>Plasmodium spp.</i> O: Discriminate between <i>P. vivax</i> , <i>P. falciparum</i> and other <i>Plasmodium spp.</i> No cross-reactivity with other pathogens	M: Discriminate <i>P. vivax</i> from other <i>Plasmodium spp.</i> O: Discriminate between <i>P. vivax</i> , <i>P. falciparum</i> and other <i>Plasmodium spp.</i> No cross-reactivity with other pathogens	M: Discriminate between <i>P. vivax, P. falciparum</i> and other <i>Plasmodium spp.</i> . O: Discriminate between all <i>Plasmodium spp.</i> No cross-reactivity with other pathogens
	Diagnostic sensitivity <sup>c</sup>	M: >95% O: ≥ 99%	M: >95% O: ≥ 99%	M: >95% O: ≥ 99%
	Diagnostic specificity <sup>c</sup>	M: >95% O: ≥ 99%	M: >95% O: ≥ 99%	M: >95% O: ≥ 99%
Operational aspects	Assay format	M and O: Single-use <i>in vitro</i> diagnostic	M and O: Single-use <i>in vitro</i> diagnostic	M: 96-well format O: 384-well format
	Assay throughput	Single assessment per test	Single assessment per test with the option to batch test up to 100 samples per run in a POC format	Batch testing in line with assay format
	Equipment	M: small (<100 cm <sup>2</sup> footprint) and portable (<5 kg) O: none	M: small (<100 cm <sup>2</sup> footprint) and portable (<5 kg) O: none	M: Transportable (<20 kg) O: Portable (<5 kg)
	Sample type	M: Capillary blood O. Capillary blood or any less invasive validated sample	M: Capillary blood O. Capillary blood or any less invasive validated sample	M: Capillary blood O. Capillary blood or any less invasive validated sample
	Sample volume (if capillary blood)	$\begin{array}{l} M: \leq 100 \ \mu L \\ O: \leq 50 \ \mu L \end{array}$	$\begin{array}{l} M: \leq 100 \; \mu L \\ O: \leq 50 \; \mu L \end{array}$	$\begin{array}{l} M: \leq 200 \; \mu L \\ O: \leq 100 \; \mu L \end{array}$
	Time-to-result	$ \begin{array}{l} M: \leq 1 \ hour \\ O: \leq 30 \ minutes \end{array} $	$\begin{array}{l} M\!: \leq \! 6 \text{ hours} \\ O\!: \leq \! 30 \text{ minutes} \end{array}$	$\begin{array}{l} M\!: \leq \! 1 \text{ month} \\ O\!: \leq \! 7 \text{ days} \end{array}$
Cost	End user price per test	M: ≤1.0 USD O: ≤0.5 USD	M: ≤2.0 USD O: ≤1.0 USD	M: ≤1.0 USD O: ≤0.1 USD
	Cost of diagnosis per sample	M: ≤2.0 USD O: ≤1.0 USD	M: ≤5.0 USD O: ≤2.0 USD	M: ≤1.2 USD O: ≤0.5 USD

#### Table 2. Summary of key distinguishing features of TPP PvA, PvB1, and PvB2

<sup>a</sup>Values in parasite per µL of blood might not be relevant for all assay types, especially for TPP PvB2, which is not for a parasitological test and includes the detection of recent infection.

<sup>b</sup>M: minimal, O: optimal

<sup>c</sup>as compared to standard PCR with a know limit of detection of 1 p/µL (PvA and PvB1) and a method with an analytical sensitivity at least equal to that of the index test (PvB2).

https://doi.org/10.1371/journal.pntd.0005516.t002

Current tests for this intended use are RDTs and microscopy and the characteristics of this TPP were established with the objective to overcome the limitations of these tests. A key performance characteristic for this TPP is the analytical sensitivity. Expert microscopy is considered to provide a LOD of 50 p/µL but this value is typically assumed to be significantly higher in many endemic areas [37,38]. A recent analysis of the analytical performances of the best-inclass Pv-pLDH RDTs indicated these would fail to detect a majority of samples containing 200 p/µL (Jimenez et al., submitted elsewhere). A minimal target LOD of 25 p/µL would therefore represent at least a two-fold improvement over the practical microscopy LOD and be a significant improvement over current RDTs. However, an optimal LOD should be equal or inferior to 5 p/ $\mu$ L, corresponding to one order of magnitude below the typical lowest peripheral parasitaemia at presentation for uncomplicated P. vivax malaria, ensuring that no clinical cases would be missed because of inadequate LOD [15,39]. Regarding diagnostic specificity, the univocal identification of *P. vivax* as the *Plasmodium* infecting species is essential as only this species and the relatively rare *P. ovale* require radical cure for liver-stage parasite removal. For areas of co-endemicity between P. vivax and P. falciparum (39 out 98 malaria endemic countries [1]), a distinguishing advantage would be the capacity to identify and discriminate between these two major species. Regarding both the diagnostic sensitivity and specificity, the minimal values have been set to at least match that of current P. falciparum RDTs and the optimal ones to provide a distinguishing advantage at 95% and 99%, respectively [25].

In terms of operational characteristics and beyond the required simplicity and rapidness of the test, stability during transport, storage and usage is important. An analysis of the typical RDT supply chain revealed that these are frequently exposed to temperature above 30°C and sometimes up to more than 40°C, hence a test destined to the same intended use needs to with-stand such harsh conditions, being ideally stable for up to 12 months at 45°C and 90% relative humidity and usable at temperatures as low as 5°C and as high as 45°C.

Another crucial element often difficult to resolve is that of cost. The cost of diagnosis (including sample collection, processing, and transmission of the results to the patient) for RDT and light microscopy were evaluated to be between 1.0 and 2.0 USD in 2011 in Uganda, a *P. falciparum* endemic country [40]. It is somewhat complex to define at what end-user price and overall diagnosis cost a PvA test might become cost-effective as the cost of misdiagnosed and relapsing *P. vivax* infections is difficult to evaluate but it was assumed that the end user price should ideally not be superior to the current price of RDT (~0.5 USD) and that the overall cost of diagnosis should not be superior to the values mentioned here above.

### TPP PvB1: Point-of-care diagnosis of sub-clinical P. vivax infection

TPP PvB1 extends the scope of PvA to address the detection of all blood-stage infections, regardless of the presence of symptoms, to enable reactive and proactive case detection and treatment (<u>S3 Table</u>). This TPP is defining the characteristics of tests that could be deployed in elimination settings to identify the asymptomatic reservoir known to contribute to residual transmission and guide blood-stage and, if appropriate, liver-stage treatments for the asymptomatic carriers.

Similar to PvA, PvB1 tests need to be deployable in a point-of-care manner (or "point-of-contact" since it would not necessarily be used in a medical care context) and therefore require very similar characteristics in terms of ease-of-training, ease-of-use, short time-to-results and operational robustness. There is no such test currently in use and while NAATs might meet many of the required characteristics, they are not easily deployable as a point-of-care diagnostic solution.

The main distinguishing feature of PvB1 as compared to PvA is the lowered analytical sensitivity needed to (i) detect a substantial fraction of the asymptomatic and low parasitaemia infections, and thus (ii) support elimination interventions by providing crucial information about these parasite populations. A modelling study investigating the case of a P. falciparum diagnostic test used to trigger focal mass drug administration (focal MDA, i.e. village-based mass drug administration in case of a local prevalence identified above a certain threshold) suggested that in such low prevalence settings, an analytical sensitivity of 20  $p/\mu L$  might suffice to ultimately reduce the parasite prevalence to zero within a ten-year time frame [41]. It is however not clear how such a model could apply to P. vivax, for which a majority of infections are relapses from liver stage parasites. A recent study evaluating the parasitaemia distribution in more than 1,500 P. vivax infected asymptomatic individuals at the Thai-Myanmar border revealed a geometric mean parasitaemia of 5.6 p/µL and a unimodal log normal distribution of parasitaemia in this population [42]. The minimally acceptable and ideal analytical sensitivity values for PvB1 tests were established around these estimates at 20 p/ $\mu$ L and 1 p/ $\mu$ L, respectively. The optimal value would allow to detect up to 58% of all asymptomatic infections according to the modelling of the data from Imwong *et al.* [42]. The other performance and operational characteristics are essentially identical between PvA and PvB1, with the notable exception that the test format of PvB1 should ideally be amenable to batch testing of up to 100 individuals relatively easily. This is in consideration of the reactive or proactive case detection scenarios for which this type of test would be used, requiring the rapid diagnosis and treatment of a potentially large number of individuals as opposed to passive case detection, where testing is normally performed on demand and as suspected cases present at health posts and medical centers.

Another key distinguishing feature of PvB1 is the cost. As mentioned above, the cost elements of diagnoses are difficult to factor in the absence of comprehensive costing analyses and cost-effectiveness studies of existing solutions. In the case of PvB1, it was assumed that the increase in analytical sensitivity requirement would translate in an increased end-user test price (minimal: 2.0 USD, optimal: 1.0 USD) and cost of diagnosis (minimal: 5.0 USD, optimal: 2.0 USD) but that these values should ideally be lower than current NAAT tests, estimated between 2 to 4 USD per reaction, excluding capital cost [43-45]. Another element typically weighted against the cost of a diagnostic test is that of a treatment course, especially when considering mass interventions. If a diagnostic test is not significantly cheaper than a treatment course, typically targeted to 1 USD or less [46], it is, from a pure economic point-of-view, cheaper to treat in the absence of testing than screen-and-treat. We dismiss this argument as too simplistic and are of the opinion that the true financial and societal costs of MDA or mass screen-and-treat cannot be distilled down to the only cost of the commodities associated with these interventions. We would not recommend for a test to be cheaper than a treatment course in order to be an adequate PvB1 diagnostic solution. This is especially true in the case of P. vivax, which requires not only blood-stage detection, but also potentially G6PD testing for radical cure.

### TPP PvB2: Population screening for P. vivax infection surveillance

The TPP PvB2 is designed to answer the needs for high quality tests for epidemiological surveillance activities (S4 Table). This TPP is similar to PvB1 in the sense that it aims to detect all infections, including asymptomatic and low parasitaemia typically not seen by RDTs or microscopy, but it differs from PvB1 in that the diagnostic outcome is not directly linked with treatment interventions at the individual level. A PvB2 test is designed to inform surveillance system and to support epidemiological surveys. Because of this nature, such a test would not need to be deployed in a point-of-care manner but would be restricted to district hospitals and national reference laboratories, and would need to provide a very high analytical sensitivity.

Current tests in this category include highly complex and specialized NAAT protocols, such as high volume quantitative PCR, reverse-transcription quantitative PCR, or PCR targeting highly repetitive elements, which all reach analytical sensitivity of approximately  $0.02 \text{ p/}\mu\text{L}$ [47-49]. The PvB2 minimal and optimal analytical sensitivity values were set at 0.1 p/µL (one order of magnitude lower than optimal PvB1 test) and 0.01 p/µL (two-fold lower than current state-of-the art technologies), respectively. Such values would in principle allow to detect up to 80% and 93% all of *P. vivax* asymptomatic infection as modelled by Imwong et al. [42]. When defining the optimal test sensitivity, it is pivotal to take into account the sampling procedures and blood sample volume. The blood volume equivalent that is added to a molecular assay critically determines the detection of low parasitaemias. All tests that target asymptomatic individuals should thus aim at maximizing the input material. For surveillance, collecting finger prick blood samples is considered feasible, whereas larger venous samples may be collected for research purposes only. In remote settings, sampling on filter paper will be required for storage and transport. This will compromise the detection of low parasitaemia in a significant way, as filter paper punches (directly added to the reaction or extracted) can only hold a limited blood volume. Ideally, 200 µL whole blood should be used for preparation of nucleic acids, and the final DNA solution should be concentrated as much as possible. Multi-copy target genes or reverse transcription reactions can help to detect a single parasite in the large blood volume sampled [48,49].

As a surveillance tool, PvB2 also includes tests that might not necessarily detect currently occurring infection but also recent past infections, such as serological tests, as a most effective way of estimating the residual transmission in an area of interest and potentially detect hypnozoite carriers. Obviously in such cases, an analytical sensitivity expressed in parasites per  $\mu$ L of blood becomes irrelevant and test-specific values would have to be defined (e.g. antibody level detected by a serology test). Ideally, the analytical specificity would also be expected to be greater than that required for PvA and PvB1 tests, and the optimal specificity would be a detection and discrimination of all five *Plasmodium spp*, infecting humans. Regarding the diagnostic sensitivity and specificity, these values were defined as similar to PvA and PvB1, however the actual reference test against which they would be determined might not easily be defined since in terms of pure analytical sensitivity, an optimal PvB2 index test is likely to be the new best standard of truth. The operational characteristics for the PvB2 TPP allow for less environment-resistant tests compared to PvA and PvB1: transport, storage and operation conditions allow for cold transportation and storage and are generally set to correspond to the typical conditions found in air-conditioned reference laboratories. The assay format is at a minimum a 96-well format and ideally a 384-well format to enable high throughput and the characterization of a high number of samples concurrently. This is linked with less stringent requirement in terms of equipment, with a tolerance for electricity requirement, up to 20 kg of equipment, and sample processing, with up to 20 steps for the assay procedure being acceptable. Similarly, the time-to-result is not of the essence in this case with an optimal time-to-results being less than 7 days, but at a minimum requirement of up to one month. These characteristics might become more critical if the test is used to target focal MDA, in which the optimal value (one week) would be a requirement.

The cost associated with such a test is arguably expected to be high, since it is likely to involve complex laboratory procedures, reagents, and equipment as well as specialized and highly trained laboratory officers. Yet, the high throughput and high volume of testing required for population surveillance interventions is expected to drive the cost of such test down to an acceptable 1.0 USD end-user price per sample (minimal requirement). The optimal value was set one order of magnitude below this, at 0.1 USD, which might be achievable for a test relying on simpler procedure and limiting the sample processing steps.

### Discussion

We present here the first consensus TPPs for P. vivax specific diagnostic tests to address the particular needs for the control and elimination of malaria due to this species. A summary of key characteristics is given in the main text (Table 2) and complete TPPs are specified in the supplementary material section (S2 to S4 Tables). Consensus-based review and discussion exercises of a large group of experts in the field guided the development of key characteristics. In this report, we describe important biological specificities of *P. vivax* and outline in a detailed manner the diagnostic needs associated with this malaria species. TPPs have been deliberately developed in a platform and technology-independent manner so that the capacity of diagnostic solutions based on any existing or new technology to fit with these profiles can be evaluated. These include (i) TPPs for the parasitological confirmation of suspected cases in clinical settings (PvA), (ii) the detection of a large number of infections, regardless of symptoms for reactive and proactive interventions (PvB1) and (iii) the identification of the majority of current or recent past infections for population-based "surveillance-response", *i.e.* an active/passive surveillance strategy that calls for response packages tailored to the respective endemic settings (PvB2). P. vivax, on average, has lower parasite densities than P. falciparum, placing an extra challenge on the analytical sensitivity requirements of *P. vivax* diagnostic tests. Another key characteristic of *P. vivax* is the existence of hypnozoites, which requires complex and potentially toxic treatment to be deployed to achieve radical cure. The haemolytic toxicity risk associated with primaquine requires a G6PD test to rule out significant deficiency in this enzyme before initiating a treatment course. G6PD testing should be an integral component of the management of P. vivax infection, and accurate tests to detect its deficiency are required. TPPs for G6PD deficiency testing have been previously defined by a group of experts and are publicly available [50]. The need for better G6PD tests has been previously recognized and a new point-of-care test is showing promising results [51-54]. With the development of improved G6PD tests, the opportunity of combination test to concomitantly detect *P. vivax* blood-stage infections and determine the G6PD status of the host should be considered from a technological and cost-effective point-of-view.

The persistence and reactivation of hypnozoites is one of the main drivers of the resilience of *P. vivax* to antimalarial interventions. Developing a specific hypnozoite test is widely regarded as an extremely challenging technical task. It would require the ability of a test to detect a very small number of metabolically inactive, or poorly active parasites, sequestered in the liver. Despite the theoretical advances of such a test, the group did not consider such a test to be crucial in achieving elimination as indirect diagnostic approaches might be more promising. This is because a P. vivax blood stage infections can be assumed with a high probability to be also associated with residual hypnozoite infections. Hence, the detection of low blood-stage parasitaemia (PvB1) together with the potential detection of past recent infections (PvB2) would appear to be a more promising avenue to identify most hypnozoite carriers. Alternatively the likelihood that hypnozoites would ultimately lead to new blood stage infections suggests that it might be sufficient to sustain diagnostic efforts to detect blood stage parasites and guide the administration of radical cure for the eventual removal of hypnozoites from all carriers. This approach would however require intense, long-lasting and costly surveillance efforts to be implemented and might require to specifically target the population most at risk of P. vivax infection.

The TPPs defined here are considered to be evolving documents by nature and should stimulate discussions within the malaria community and be periodically revised to ensure that they still represent the actual needs for the diagnosis of *P. vivax*. These profiles will also serve to evaluate current and future tests, and should facilitate an ongoing discussion on the capacity of these tests to address the existing *P. vivax* diagnostics gaps. They will also serve to guide innovation and development efforts in areas of greatest need. The first one of these being better tests for the clinical management of *P. vivax* infections, so that the positive impact that has been observed for affordable high-quality *P. falciparum* RDTs can also be obtained for *P. vivax* malaria. Second, highly sensitive point-of-care tests will be needed to support reactive and proactive infection detection strategies through the accurate detection of asymptomatic and low density parasitaemia for those settings where plain MDA will not be adequate, that is in areas of low prevalence, where overtreatment would be unacceptably high, or in areas with a high prevalence of G6PD deficiency. Finally, no control or elimination programme can bypass the need for high quality and highly sensitive prevalence survey data. While high-performance but also high-cost and high technology NAAT approaches exist, cheaper and more standardized solutions are needed.

The limitations of the TPPs are directly linked with those of our knowledge on *P. vivax* malaria. For instance, more research is clearly needed on the epidemiology of asymptomatic *P. vivax* infections. While this parasite population has been described in South East-Asia [42], similar quantitative investigations are lacking in other settings. Similarly, a better understanding of the prevalence and distribution of *P. vivax* in sub-Saharan Africa [55], a place from which this species was historically thought to be mostly absent, is needed. Research is also needed for the potential identification of serological biomarker which could facilitate the detection of recent past infections. In addition, cost-effectiveness studies of current diagnostics and interventions are required to guide improvements and definition of parameter to measure impact.

Since more international organizations are becoming aware of the significant and specific role played by *P. vivax* malaria, we think that this initial set of *P. vivax* specific diagnostic TPPs highlights particular diagnostic gaps and defines specific parameters that may guide the global health community to address these needs coherently and thus contribute to the ongoing successful control and elimination efforts.

#### Disclaimer

The views and opinions expressed in this article are consensus views and opinions from all individual authors and do not necessarily reflect the official policy or position of any of the authors institutions.

#### Supporting information

**S1 Table. Summary of existing malaria diagnostic TPPs.** (PDF)

**S2 Table.** TPP PvA: Diagnosis of *Plasmodium vivax* malaria acute infection. (PDF)

**S3 Table.** TPP PvB1: Point-of-care diagnosis of subpatent *Plasmodium vivax* infection. (PDF)

**S4 Table. TPP PvB2: Population screening for** *Plasmodium vivax* **infection surveillance.** (PDF)

### **Author Contributions**

**Conceptualization:** XCD MPA JKB QC JC MD CD IF DG MH SH NL AM IM JS AR JR MT IJG.

Funding acquisition: IJG.

Investigation: XCD MPA JKB QC JC MD CD IF DG MH SH NL AM IM JS AR JR MT IJG.

Methodology: XCD IJG.

Project administration: XCD.

Supervision: IJG.

Writing - original draft: XCD.

Writing – review & editing: XCD MPA JKB QC JC MD CD IF DG MH SH NL AM IM JS AR JR MT IJG.

#### References

- 1. World Health Organization. World Malaria Report 2015. 2015.
- Gething PW, Elyazar IRF, Moyes CL, Smith DL, Battle KE, Guerra CA, et al. A Long Neglected World Malaria Map: Plasmodium vivax Endemicity in 2010. Carlton JM, editor. PLoS Negl Trop Dis. Public Library of Science; 2012; 6: e1814. https://doi.org/10.1371/journal.pntd.0001814 PMID: 22970336
- **3.** World Health Organization. Control and Elimination of Plasmodium Vivax Malaria. 2015.
- 4. Malaria-Eliminating Countries | Malaria Elimination Group [Internet]. [cited 16 Jun 2015]. Available: http://www.malariaeliminationgroup.org/resources/elimination-countries
- Kevin Baird J. Malaria caused by Plasmodium vivax: recurrent, difficult to treat, disabling, and threatening to life—the infectious bite preempts these hazards. Pathog Glob Health. 2013; 107: 475–479. https://doi.org/10.1179/2047772413Z.00000000179 PMID: 24428831
- Robinson LJ, Wampfler R, Betuela I, Karl S, White MT, Suen CSNLW, et al. Strategies for Understanding and Reducing the Plasmodium vivax and Plasmodium ovale Hypnozoite Reservoir in Papua New Guinean Children: A Randomised Placebo-Controlled Trial and Mathematical Model. Seidlein von L, editor. PLoS Med. Public Library of Science; 2015; 12: e1001891. <u>https://doi.org/10.1371/journal.pmed.</u> 1001891 PMID: 26505753
- Adekunle AI, Pinkevych M, McGready R, Luxemburger C, White LJ, Nosten F, et al. Modeling the dynamics of Plasmodium vivax infection and hypnozoite reactivation in vivo. PLoS Negl Trop Dis. Public Library of Science; 2015; 9: e0003595. https://doi.org/10.1371/journal.pntd.0003595 PMID: 25780913
- Eichner M, Diebner HH, Molineaux L, Collins WE, Jeffery GM, Dietz K. Genesis, sequestration and survival of Plasmodium falciparum gametocytes: parameter estimates from fitting a model to malariatherapy data. Trans R Soc Trop Med Hyg. Oxford University Press; 2001; 95: 497–501. PMID: 11706658
- Pukrittayakamee S, Imwong M, Singhasivanon P, Stepniewska K, Day NJ, White NJ. Effects of Different Antimalarial Drugs on Gametocyte Carriage in P. Vivax Malaria. Am J Trop Med Hyg. American Society of Tropical Medicine and Hygiene; 2008; 79: 378–384. PMID: <u>18784229</u>
- Taylor LH, Read AF. Why so few transmission stages? Reproductive restraint by malaria parasites. Parasitol Today (Regul Ed). Elsevier; 1997; 13: 135–140.
- 11. Bousema T, Drakeley C. Epidemiology and infectivity of Plasmodium falciparum and Plasmodium vivax gametocytes in relation to malaria control and elimination. Clin Microbiol Rev. 2011; 24: 377–410. https://doi.org/10.1128/CMR.00051-10 PMID: 21482730
- Malleret B, Li A, Zhang R, Tan KSW, Suwanarusk R, Claser C, et al. Plasmodium vivax: restricted tropism and rapid remodeling of CD71-positive reticulocytes. Blood. 2015; 125: 1314–1324. https://doi. org/10.1182/blood-2014-08-596015 PMID: 25414440
- Baird JK. Evidence and implications of mortality associated with acute Plasmodium vivax malaria. Clin Microbiol Rev. 2013; 26: 36–57. https://doi.org/10.1128/CMR.00074-12 PMID: 23297258
- 14. McKenzie FE, Jeffery GM, Collins WE. Plasmodium vivax blood-stage dynamics. J Parasitol. 2002; 88: 521–535. https://doi.org/10.1645/0022-3395(2002)088[0521:PVBSD]2.0.CO;2 PMID: 12099421
- McKenzie FE, Sirichaisinthop J, MILLER RS, Gasser RA, Wongsrichanalai C. Dependence of malaria detection and species diagnosis by microscopy on parasite density. Am J Trop Med Hyg. American Society of Tropical Medicine and Hygiene; 2003; 69: 372–376. PMID: 14640495
- McKenzie FE, Jeffery GM, Collins WE. Gametocytemia and fever in human malaria infections. J Parasitol. 2007; 93: 627–633. https://doi.org/10.1645/GE-1052R.1 PMID: 17626355
- 17. Barber BE, William T, Grigg MJ, Parameswaran U, Piera KA, Price RN, et al. Parasite Biomass-Related Inflammation, Endothelial Activation, Microvascular Dysfunction and Disease Severity in Vivax Malaria.

Stevenson MM, editor. PLoS Pathog. Public Library of Science; 2015; 11: e1004558. https://doi.org/10. 1371/journal.ppat.1004558 PMID: 25569250

- Anstey NM, Douglas NM, Poespoprodjo JR, Price RN. Plasmodium vivax: Clinical Spectrum, Risk Factors and Pathogenesis. BS:APAR. Elsevier; 2012. pp. 151–201.
- Cheng Q, Cunningham J, Gatton ML. Systematic review of sub-microscopic P. vivax infections: prevalence and determining factors. Rodrigues MM, editor. PLoS Negl Trop Dis. Public Library of Science; 2015; 9: e3413. https://doi.org/10.1371/journal.pntd.0003413 PMID: 25569135
- Howes RE, Piel FB, Patil AP, Nyangiri OA, Gething PW, Dewi M, et al. G6PD Deficiency Prevalence and Estimates of Affected Populations in Malaria Endemic Countries: A Geostatistical Model-Based Map. Seidlein von L, editor. PLoS Med. Public Library of Science; 2012; 9: e1001339. <u>https://doi.org/10.1371/journal.pmed.1001339</u> PMID: 23152723
- 21. Ingram RJH, Crenna-Darusallam C, Soebianto S, Noviyanti R, Baird JK. The clinical and public health problem of relapse despite primaquine therapy: case review of repeated relapses of Plasmodium vivax acquired in Papua New Guinea. Malar J. BioMed Central; 2014; 13: 1.
- Bennett JW, Pybus BS, Yadava A, Tosh D, Sousa JC, McCarthy WF, et al. Primaquine failure and cytochrome P-450 2D6 in Plasmodium vivax malaria. N Engl J Med. 2013; 369: 1381–1382. https://doi.org/ 10.1056/NEJMc1301936 PMID: 24088113
- Baird JK, Hoffman SL. Primaquine therapy for malaria. Clin Infect Dis. 2004; 39: 1336–1345. <u>https://doi.org/10.1086/424663</u> PMID: 15494911
- Maltha J, Gillet P, Cnops L, van den Ende J, van Esbroeck M, Jacobs J. Malaria rapid diagnostic tests: Plasmodium falciparum infections with high parasite densities may generate false positive Plasmodium vivax pLDH lines. Malar J. 2010; 9: 198. https://doi.org/10.1186/1475-2875-9-198 PMID: 20618990
- Abba K, Kirkham AJ, Olliaro PL, Deeks JJ, Donegan S, Garner P, et al. Rapid diagnostic tests for diagnosing uncomplicated non-falciparum or Plasmodium vivax malaria in endemic countries. Abba K, editor. Cochrane Database Syst Rev. Chichester, UK: John Wiley & Sons, Ltd; 2014; 12: CD011431.
- Jang JW, Cho CH, Han E-T, An SSA, Lim CS. pLDH level of clinically isolated Plasmodium vivax and detection limit of pLDH based malaria rapid diagnostic test. Malar J. 2013; 12: 181. <u>https://doi.org/10. 1186/1475-2875-12-181 PMID: 23731660</u>
- Mouatcho JC, Goldring JPD. Malaria rapid diagnostic tests: challenges and prospects. J Med Microbiol. 2013; 62: 1491–1505. https://doi.org/10.1099/jmm.0.052506-0 PMID: 24048274
- The malERA Consultative Group on Diagnoses and Diagnostics. A Research Agenda for Malaria Eradication: Diagnoses and Diagnostics. PLoS Med. 2011; 8: e1000396. https://doi.org/10.1371/journal. pmed.1000396 PMID: 21311583
- World Health Organization. Malaria rapid diagnostic test performance: results of WHO product testing of malaria RDTs: round 6 (2014–2015). 2015;: 1–154.
- FIND—Malaria RDT product testing: interactive guide. In: rdt-interactive-guide.org [Internet]. [cited 10 Jun 2016]. Available: http://www.rdt-interactive-guide.org/
- Mekonnen SK, Aseffa A, Medhin G, Berhe N, Velavan TP. Re-evaluation of microscopy confirmed Plasmodium falciparum and Plasmodium vivax malaria by nested PCR detection in southern Ethiopia. Malar J. BioMed Central; 2014; 13: 1.
- 32. Barber BE, William T, Grigg MJ, Yeo TW, Anstey NM. Limitations of microscopy to differentiate Plasmodium species in a region co-endemic for Plasmodium falciparum, Plasmodium vivax and Plasmodium knowlesi. Malar J. BioMed Central; 2013; 12: 1.
- WHO | WHO policy recommendation on malaria diagnostics in low transmission settings. In: WHO [Internet]. World Health Organization; [cited 16 Jun 2015]. Available: <u>http://www.who.int/malaria/</u> publications/atoz/diagnotics-low-transmission-settings/en/
- World Health Organization. WHO Evidence Review Group on Malaria Diagnosis in Low Transmission Settings [Internet]. 21 Mar 2014 [cited 11 Oct 2015] pp. 1–33. Available: http://www.who.int/malaria/ mpac/mpac\_mar2014\_diagnosis\_low\_transmission\_settings\_report.pdf
- **35.** PATH. Target Product Profile: Point-of-Care Malaria Infection Detection Test. In: sites.path.org [Internet]. [cited 25 Jun 2016]. Available: http://sites.path.org/dx/files/2012/11/DIAMETER\_IDT\_TPP\_ FINAL\_forwebsite.pdf
- Tietje K, Hawkins K, Clerk C, Ebels K, McGray S, Crudder C, et al. The essential role of infection-detection technologies for malaria elimination and eradication. Trends Parasitol. Elsevier; 2014; 30: 259– 266. https://doi.org/10.1016/j.pt.2014.03.003 PMID: 24726857
- Mukadi P, Gillet P, Lukuka A, Atua Ben, Kahodi S, Lokombe J, et al. External quality assessment of malaria microscopy in the Democratic Republic of the Congo. Malar J. BioMed Central; 2011; 10: 1.

- Joanny F, Löhr SJ, Engleitner T, Lell B, Mordmüller B. Limit of blank and limit of detection of Plasmodium falciparum thick blood smear microscopy in a routine setting in Central Africa. Malar J. BioMed Central; 2014; 13: 1.
- Anstey NM, Douglas NM, Poespoprodjo JR, Price RN. Plasmodium vivax: clinical spectrum, risk factors and pathogenesis. Adv Parasitol. Elsevier; 2012; 80: 151–201. <u>https://doi.org/10.1016/B978-0-12-397900-1.00003-7 PMID: 23199488</u>
- Batwala V, Magnussen P, Hansen KS, Nuwaha F. Cost-effectiveness of malaria microscopy and rapid diagnostic tests versus presumptive diagnosis: implications for malaria control in Uganda. Malar J. BioMed Central Ltd; 2011; 10: 372. https://doi.org/10.1186/1475-2875-10-372 PMID: 22182735
- Slater HC, Ross A, Ouédraogo AL, White LJ, Nguon C, Walker PGT, et al. Assessing the impact of next-generation rapid diagnostic tests on Plasmodium falciparum malaria elimination strategies. Nature. Nature Publishing Group; 2015; 528: S94–S101. https://doi.org/10.1038/nature16040 PMID: 26633771
- Imwong M, Stepniewska K, Tripura R, Peto TJ, Lwin KM, Vihokhern B, et al. Numerical Distributions of Parasite Densities During Asymptomatic Malaria. J INFECT DIS. 2015.
- 43. Khan SA, Ahmed S, Mushahid N, Anwer M, Saeed S, Khan FA, et al. Comparison of real time polymerase chain reaction with microscopy and antigen detection assay for the diagnosis of malaria. J Coll Physicians Surg Pak. 2013; 23: 787–792. https://doi.org/11.2013/JCPSP.787792 PMID: 24169386
- 44. Canier L, Khim N, Kim S, Sluydts V, Heng S, Dourng D, et al. An innovative tool for moving malaria PCR detection of parasite reservoir into the field. Malar J. BioMed Central; 2013; 12: 1.
- 45. Erdman LK, Kain KC. Molecular diagnostic and surveillance tools for global malaria control. Travel Med Infect Dis. 2008; 6: 82–99. https://doi.org/10.1016/j.tmaid.2007.10.001 PMID: 18342279
- Burrows JN, van Huijsduijnen RH, Möhrle JJ, Oeuvray C, Wells TN. Designing the next generation of medicines for malaria control and eradication. Malar J. BioMed Central; 2013; 12: 187. https://doi.org/ 10.1186/1475-2875-12-187 PMID: 23742293
- Imwong M, Hanchana S, Malleret B, Rénia L, Day NPJ, Dondorp A, et al. High-Throughput Ultrasensitive Molecular Techniques for Quantifying Low-Density Malaria Parasitemias. Gilligan PH, editor. J Clin Microbiol. American Society for Microbiology; 2014; 52: 3303–3309. <u>https://doi.org/10.1128/JCM.</u> 01057-14 PMID: 24989601
- Hofmann N, Mwingira F, Shekalaghe S, Robinson LJ, Mueller I, Felger I. Ultra-Sensitive Detection of Plasmodium falciparum by Amplification of Multi-Copy Subtelomeric Targets. Seidlein von L, editor. PLoS Med. Public Library of Science; 2015; 12: e1001788. https://doi.org/10.1371/journal.pmed. 1001788 PMID: 25734259
- 49. Murphy SC, Prentice JL, Williamson K, Wallis CK, Fang FC, Fried M, et al. Real-time quantitative reverse transcription PCR for monitoring of blood-stage Plasmodium falciparum infections in malaria human challenge trials. Am J Trop Med Hyg. American Society of Tropical Medicine and Hygiene; 2012; 86: 383–394. https://doi.org/10.4269/ajtmh.2012.10-0658 PMID: 22403305
- Domingo GJ, Satyagraha AW, Anvikar A, Baird K, Bancone G, Bansil P, et al. G6PD testing in support of treatment and elimination of malaria: recommendations for evaluation of G6PD tests. Malar J. BioMed Central; 2013; 12: 1.
- Oo NN, Bancone G, Maw LZ, Chowwiwat N, Bansil P, Domingo GJ, et al. Validation of G6PD Point-of-Care Tests among Healthy Volunteers in Yangon, Myanmar. PLoS ONE. Public Library of Science; 2016; 11: e0152304. https://doi.org/10.1371/journal.pone.0152304 PMID: 27035821
- 52. Fricken von ME, Weppelmann TA, Eaton WT, Masse R, Beau de Rochars MVE, Okech BA. Performance of the CareStart glucose-6-phosphate dehydrogenase (G6PD) rapid diagnostic test in Gressier, Haiti. Am J Trop Med Hyg. American Society of Tropical Medicine and Hygiene; 2014; 91: 77–80. https://doi.org/10.4269/ajtmh.14-0100 PMID: 24778197
- Satyagraha AW, Sadhewa A, Elvira R, Elyazar I, Feriandika D, Antonjaya U, et al. Assessment of Pointof-Care Diagnostics for G6PD Deficiency in Malaria Endemic Rural Eastern Indonesia. Sinnis P, editor. PLoS Negl Trop Dis. Public Library of Science; 2016; 10: e0004457. https://doi.org/10.1371/journal. pntd.0004457 PMID: 26894297
- 54. Espino FE, Bibit J-A, Sornillo JB, Tan A, Seidlein von L, Ley B. Comparison of Three Screening Test Kits for G6PD Enzyme Deficiency: Implications for Its Use in the Radical Cure of Vivax Malaria in Remote and Resource-Poor Areas in the Philippines. Rénia L, editor. PLoS ONE. Public Library of Science; 2016; 11: e0148172. https://doi.org/10.1371/journal.pone.0148172 PMID: 26849445
- 55. Howes RE, Reiner RC Jr, Battle KE, Longbottom J, Mappin B, Ordanovich D, et al. Plasmodium vivax Transmission in Africa. del Portillo HA, editor. PLoS Negl Trop Dis. Public Library of Science; 2015; 9: e0004222. https://doi.org/10.1371/journal.pntd.0004222 PMID: 26587988